Author’s response to reviews

Title: Atypical Presentation of Hemorrhagic Shock in Pregnancy: A Case Highlighting the Developing Field of Emergency Medicine in Israel

Authors:

Baruch Berzon (baruchbe@assuta.co.il)

Michael Gleenberg (mgleen@gmail.com)

Joseph Offenbacher (joffenba@montefiore.org)

Debra West (west_debra@walla.co.il)

Version: 2 Date: 08 Sep 2019

Author’s response to reviews:

Please see attached document.